Author's response to reviews

Title: Locally relapsed and metastatic uterine leiomyoma: a case report.

Authors:

Ambrogio P Londero (ambrogio.londero@gmail.com)
Patrizia Perego (patrizia.perego@hsgerardo.org)
Costantino Mangioni (c.mangioni@hsgerardo.org)
Ralph J Lelle’ (info@lellenet.de)
Franco Londero (londero.franco@libero.it)
Diego Marchesoni (diego.marchesoni@uniud.it)

Version: 2 Date: 4 December 2007

Author's response to reviews:

We have revised the spelling, the English grammar, and we have modified the other things according to your suggestions and to authors instructions. Then we ask an English mother-tongue to proof read our manuscript. We apologize for our English that seems more "pigeon English" than English. And we hope you could understand what we would like to express with this article.

Thanks for your collaboration.
Sincerely
ambrogio p londero